Author's response to reviews

Title: Cholesteatoma of the external ear canal: etiological factors, symptoms and clinical findings in a series of 48 cases

Authors:

Hanne H Owen (hhowen@dadlnet.dk)
Jorn Rosborg (jros@gh.gl)
Michael Gaihede (mgaihede@aas.nja.dk)

Version: 4 Date: 21 December 2006

Author’s response to reviews:

To The BioMed Central Editorial Production Team
Thank you for your suggestions. We have now encorporated the formatting changes requested in our manuscript.
Yours sincerely
Hanne Owen